Author's response to reviews

Title: Management of major bile duct injury after laparoscopic cholecystectomy: a case report

Authors:
Andreas Manouras (amanouras@hippocratios.gr)
Nikolaos Pararas (npararas@hotmail.com)
Pantelis Antonakis (antonakispantelis@yahoo.gr)
Emannuel E. Lagoudiannakis (redemlag@yahoo.gr)
George Papageorgiou (awace@yahoo.gr)
Ioannis G. Dalianoudis (johndalas@gmail.com)
Manoussos M. Konstadoulakis (lbsures@hippocratios.gr)
Vaggelogianniss Katergiannakis (vkater@hotmail.com)

Version: 7 Date: 29 October 2008

Author's response to reviews:

We have accept the changes that you have proposed. We have correct the mistake in reference number 7. After authors meeting, authors IGD and MMK should be given to them author accreditation as they have played an important role to the manuscript. Authors agree that VK should not be included to the authors thus he will be mentioned to the acknowledgement section.

Thank you